Dear Editor:

We are resubmitting a second revised paper entitled "Mobile Phone Intervention for Increasing Adherence to Treatment for Type 2 Diabetes in an Urban Area of Bangladesh: Protocol for a Randomized Controlled Trial" according to the editor’s suggestion. We have followed the reviewers’ comments to adjust or correct our manuscript. The following pages are our replies to reviewers’ comments.

We would like to extend our thanks and gratitude to the reviewers who have provided with detailed feedback and suggestion to improve our manuscript.

Our manuscript has not previously been published in print or in electronic format and has not under consideration elsewhere. Besides, none of the authors has any conflict of interest in connection with this paper. We hope that you will consider our article for publication and look forward to hearing from you.

Sincerely yours,

Dr. Sheikh Mohammed Shariful Islam, MBBS, MPH
Senior Research Investigator, Center for Control of Chronic Diseases (CCCD), ICDDR, B, 68, Shaheed Tajuddin Ahmed Sarani, Mohakhali, Dhaka-1212, Bangladesh
Phone: +88-01939366930
Email: shariful.islam@icddrb.org